Assessing the Impact of Misclassification Error on an Epidemiological Association between Two Helminthic Infections

Mushfiqur R. Tarafder^{1,2*}, Hélène Carabin¹, Stephen T. McGarvey³, Lawrence Joseph⁴, Ernesto Balolong Jr.⁵, Remigio Olveda⁵

1 Department of Biostatistics and Epidemiology, College of Public Health, University of Oklahoma Health Sciences Center, Oklahoma City, Oklahoma, United States of America, 2 Department of Family, Community and Rural Health, The Commonwealth Medical College, Scranton, Pennsylvania, United States of America, 3 International Health Institute, Brown University, Providence, Rhode Island, United States of America, 4 Division of Clinical Epidemiology, McGill University Health Centre, Montréal, Canada, 5 Research Institute for Tropical Medicine, Alabang, Muntinlupa City, Philippines

Abstract

Background: Polyparasitism can lead to severe disability in endemic populations. Yet, the association between soil-transmitted helminth (STH) and the cumulative incidence of *Schistosoma japonicum* infection has not been described. The aim of this work was to quantify the effect of misclassification error, which occurs when less than 100% accurate tests are used, in STH and *S. japonicum* infection status on the estimation of this association.

Methodology/Principal Findings: Longitudinal data from 2276 participants in 50 villages in Samar province, Philippines treated at baseline for *S. japonicum* infection and followed for one year, served as the basis for this analysis. Participants provided 1–3 stool samples at baseline and 12 months later (2004–2005) to detect infections with STH and *S. japonicum* using the Kato-Katz technique. Variation from day-to-day in the excretion of eggs in feces introduces individual variations in the sensitivity and specificity of the Kato-Katz to detect infection. Bayesian logit models were used to take this variation into account and to investigate the impact of misclassification error on the association between these infections. Uniform priors for sensitivity and specificity of the diagnostic test to detect the three STH and *S. japonicum* were used. All results were adjusted for age, sex, occupation, and village-level clustering. Without correction for misclassification error, the odds ratios (ORs) between hookworm, *Ascaris lumbricoides*, and *Trichuris trichiura*, and *S. japonicum* infections were 1.28 (95% Bayesian credible intervals: 0.93, 1.76), 0.91 (95% BCI: 0.66, 1.26), and 1.11 (95% BCI: 0.80, 1.55), respectively, and 2.13 (95% BCI: 1.16, 4.08), 0.74 (95% BCI: 0.43, 1.25), and 1.32 (95% BCI: 0.80, 2.27), respectively, after correction for misclassification error for both exposure and outcome.

Conclusions/Significance: The misclassification bias increased with decreasing test accuracy. Hookworm infection was found to be associated with increased 12-month cumulative incidence of *S. japonicum* infection after correction for misclassification error. Such important associations might be missed in analyses which do not adjust for misclassification errors.

Citation: Tarafder MR, Carabin H, McGarvey ST, Joseph L, Balolong E Jr, et al. (2011) Assessing the Impact of Misclassification Error on an Epidemiological Association between Two Helminthic Infections. PLoS Negl Trop Dis 5(3): e995. doi:10.1371/journal.pntd.0000995

Editor: Jeffrey Michael Bethony, George Washington University, United States of America

Received May 5, 2010; Accepted March 3, 2011; Published March 29, 2011

Copyright: © 2011 Tarafder et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: This work was supported by the National Institutes of Health/National Science Foundation Ecology of Infectious Diseases program [R01 TW01582]. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* E-mail: mtarafder@tcmedc.org

Introduction

Polyparasitism is a common feature in parasite endemic regions, which includes most developing countries [1,2]. High prevalence of co-infection with soil-transmitted helminths (STHs), which include roundworm (*Ascaris lumbricoides*), whipworm (*Trichuris trichiura*), and hookworm (*Ancylostoma duodenale* and *Necator americanus*), and *Schistosoma* spp. has been reported [3,4]. Together, these infections correspond to an estimated 43.5 million disability-adjusted life years (DALYs) lost annually [5,6]. Schistosomiasis and STH infections are associated with conditions of poverty, such as poor hygiene, lack of safe water, inadequate sanitation and factors such as water management systems, age, gender, and farming related activities [4,5,7–14].

Laboratory studies suggest that infection with one helminth may influence the outcome of infection with another helminth [15]. Positive cross-sectional correlation and synergism between schistosome and STH infections have been reported [2,3,6,16–18]. Immunosuppressive effect of STH has been reported, particularly with hookworm infections [19,20]. The influence of STH infection on risk of infection with schistosomes has not been epidemiologically investigated. One challenge faced by investigators is the use of a less than perfect diagnostic test. The outcome, exposures, confounding variables, or any combination of these can contain errors [21–23]. Error in identification of infection status occurs when the test used to identify the infection is not 100% accurate, or not a 'gold' standard test [21,24,25].

Author Summary

Hookworm, roundworm, and whipworm are collectively known as soil-transmitted helminths. These worms are prevalent in most of the developing countries along with another parasitic infection called schistosomiasis. The tests commonly used to detect infection with these worms are less than 100% accurate. This leads to misclassification of infection status since these tests cannot always correctly indentify infection. We conducted an epidemiological study where such a test, the Kato-Katz technique, was used. In our study we tried to show how misclassification error can influence the association between soil-transmitted helminth infection and schistosomiasis in humans. We used a statistical technique to calculate epidemiological measures of association after correcting for the inaccuracy of the test. Our results show that there is a major difference between epidemiological measures of association before and after the correction of the inaccuracy of the test. After correction of the inaccuracy of the test, soiltransmitted helminth infection was found to be associated with increased risk of acquiring schistosomiasis. This has major public health implications since effective control of one worm can lead to reduction in the occurrence of another and help to reduce the overall burden of worm infection in affected regions.

Schistosoma japonicum and STH infections are most commonly detected by examining a stool sample under the microscope for the presence of parasitic eggs. Variation from day-to-day in the excretion of *S. japonicum* and STH eggs in human feces has been reported [26–29]. Collecting stool samples over consecutive days has been shown to improve the sensitivity of coprological tests like Kato-Katz [29,30]. However, in practice, an unequal number of stool specimens per subject are collected as it is difficult to collect the desired number of stool samples from each subject. This produces potential complications in diagnosing *S. japonicum* and STH infections as the sensitivity and specificity of the diagnostic tests vary according to the number of stool samples examined [31,32].

The purpose of this study was to show the impact of adjusting for misclassification error in estimating the effect of STH infections on the 12-months cumulative incidence of *S. japonicum* infection. Measuring such impact will contribute to a better understanding of the association between STH and schistosomiasis.

Materials and Methods

Ethics statement

The research was approved by the institutional review board (IRB) of the Brown University in the United States and the IRB of the Research Institute for Tropical Medicine in the Philippines. The data analysis component of the study was reviewed and approved by the University of Oklahoma Health Sciences Center IRB. The chiefs of all villages were asked permission for the village to be included in the study. In addition, all eligible participants were asked for their consent to participate. Only those individuals who provided written informed consent were included. Written informed consent for individuals below 18 years old was obtained and provided by parents or legal guardians.

Source of data

We used data from a longitudinal study conducted between January 2004 and December 2005 in the province of Samar, the Philippines. The main purpose of the original study was to assess the effect of water and animal management systems on the transmission of *S. japonicum* infection. The design of the baseline study was described elsewhere [33]. A brief summary is given below.

Study population

Seventy-five out of 134 villages endemic for *S. japonicum* in Samar in 2002 were eligible for participation [33]. The inclusion criteria were safety and accessibility of the field team, location and number of households in each village. Twenty-five primarily rainfed villages and 25 villages with some form of man-made irrigation system were selected.

Eligible households were those of at least five members and where at least one member was working full time in a rain-fed farm in "rain-fed" villages and at least 50% of the time in a manmade irrigated farm in "irrigated" villages. A maximum of 35 eligible households were randomly selected from each village using the following procedure. A list of 50 random numbers was created (one list per village). Eligible households were allocated consecutive numbers and visited in the order chosen at random. If a household refused to participate, the next available household was asked to participate. When 35 or fewer households were eligible in a village, they were all invited to participate in the study. At most six individuals including at least one full-time rice farmer were selected at random from each household.

Baseline data and stool collection

An individual-level interview included questions on age, gender, and occupation. Participants were asked to provide one stool sample (morning or first) per day for three consecutive days. Each participant provided between one and three stool samples. If a participant provided a stool sample on one of the three days but was unable for any reason to provide stool samples on other days, that person was still considered as a stool sample provider. Stool envelopes (of wax paper and book paper) with popsicle sticks were distributed to participants a day before the actual stool collection. At least thumb-size stool samples were submitted. Portions from different parts of the stool were taken to fill up the template. Although consistency of the stool sample was not recorded, only pasty to formed stool could be accommodated in the stool envelopes. Stool samples were processed 2-3 h after collection. Two slides were prepared from each stool sample. All slides were placed in a styrofoam box with cold packs inside at the end of each collection day. At the end of each collection week all slides were brought to a designated laboratory and transferred to a refrigerator. The time delay between stool sample processing and microscopic reading associated with day one stool collection (provided by 99.45% of participants) ranged from less than 24 hours to as long as 20 days with a median of 4 days (interquartile range: 2-6 days). Stool samples were examined for the presence of eggs of S. japonicum and the three STHs. No distinction between N. americanus and A. duodenale eggs was made, although prior reports from the Philippines found exclusively Necator spp. infections [34]. The Kato-Katz technique was used to detect the helminth eggs in stool samples [35]. The number of eggs per gram of stool (epg) was counted for S. japonicum. Although the eggs of each of the STHs were originally documented qualitatively in five response categories (0, + through ++++), STHs were considered as dichotomous variables (observed infected or uninfected) since the researchers were particularly interested in this association. Also, since the infection of interest of the original study was schistosomiasis, the semi-quantitative ascertainment of STH infection may not have been as accurate as that for schistosomiasis. Laboratory technicians were blinded to the identity of the provider of the stool sample they were preparing and reading and did not

know if two stool samples were from the same participant (two consecutive day's sample).

Mass treatment

Details about the mass treatment have been published elsewhere [36]. Briefly, following the baseline data and stool collection, all residents who were ≥ 5 years of age at the time and living in the 50 study villages were offered praziquantel. Praziquantel was administered in two equal split doses to give each individual a total of 60 mg/kg. The split doses were administered 4 hours apart with the first dose usually between 9 am and noon. All participants who provided baseline stool samples had been notified of their test results before treatment was offered. Before mass treatment, community preparation was implemented and an effort was made to ensure all cases found to be positive for S. japonicum were treated. Despite these efforts, the village-level participation proportion varied from 16% to 81% [36]. The parasitological test results were shared with the local ministry of health and the national schistosomiasis control team and it was decided to treat villagers positive to STH at the end of the whole study, that is, after the 12-months follow-up. This approach was approved by both IRBs.

Follow-up stool sample

All of the study participants were asked to provide three stool samples over three consecutive days 12 months after the mass treatment. All individuals who provided at least one stool sample were considered as follow-up stool sample providers. Stool samples were processed and examined in the same manner and by the same people as at baseline.

Statistical analysis

Some of the participants who provided the baseline stool samples did not participate in the mass treatment program. Moreover, not all participants provided stool samples during the follow-up survey. The 12-month cumulative incidence of *S. japonicum* infection/reinfection following mass treatment can only be calculated among the "at-risk" participants who provided at least one stool sample at baseline and follow-up and received treatment. For the purpose of this study, we assumed 100% efficacy of praziquantel for the treatment of schistosomiasis.

As mentioned earlier, we obtained between one and three stool samples on consecutive days from each participant at baseline and follow-up. This introduces individual variations in the sensitivity and specificity of the Kato-Katz to detect infection. To take this variation into account, and to adjust for the village-level clustering of infection, we used a Bayesian latent class hierarchical cumulative-logit regression model based on a method described by Joseph and others (1995) and adapted to our problem (1, 2, or 3 days of sampling) for *S. japonicum* in animals and in humans in the Philippines [25,33,37,38].

The probability of any single test being positive is the sum of the probability of a true positive result and the probability of a false positive result. If P is the total probability of a positive test, then, from the properties of diagnostic tests, we have

P = probability of true positive + probability of false positive

- = [prevalence * sensitivity] +
 - [(1 prevalence) * (1 specificity)].

When there is more than one test per person, the properties of multiple tests can be modeled using probability P as the probability

parameter of a binomial distribution, assuming that the tests are independent from each other [37]. In the absence of a 'gold' standard test, the true status of each subject is unknown, and hence can be considered as 'latent data'. According to Bayes' theorem, the joint posterior distribution is proportional to the product of the likelihood function and prior distribution, from which all inferences can be obtained. The posterior distribution is not directly available, but inferences about each parameter are available using a Gibbs sampler algorithm, as has become standard in Bayesian analysis. The unknown true infection status for each subject can be estimated once the sensitivity and specificity have been estimated.

The main outcome of interest here is the probability distribution of the true *S. japonicum* infection category at follow-up. *S. japonicum* epg counts were grouped into three categories namely: uninfected (0 epg), light infection (1 to 100 epg) and moderate to heavy infection (over 100 epg) [33]. With a three-category outcome variable, classification errors must be further subdivided. For example, when a participant who is truly negative tests positive, there are two possible errors and 1specificity or the false positive rate must be divided into light or moderate/heavy misclassification errors. The exposure of interest is the probability distribution of true STH infection status (for a particular STH) classified as positive or negative. Separate models were carried out for each of the three STHs.

Each hierarchical model consists of three levels, as follows: the first level includes one intercept parameter for each village and independent variables for age, sex, occupation, and one of the STHs under study. At the second level of the hierarchical model, the intercept parameters from each of the 50 villages are modeled as a linear regression to account for the clustering of infection within village. At the third level, prior distributions were specified for all parameters. Uniform (uninformative) prior distributions on the range from 0 to 1 (parameters of the beta distribution: $\alpha = 1$, $\beta = 1$) were used for sensitivity and specificity of all three STH infections. For *S. japonicum*, prior specificity mean (SD) for one stool sample was based on our previous work and set to 94.7% (4.0%), and prior sensitivities (SD) for detecting light infection and moderately to heavy infection were set to 54.1% (10.1%) and 75.3% (15%), respectively [33].

The above model was modified to construct three additional models: one model accounted for misclassification error in outcome but not in exposure, one accounted for misclassification error in exposure but not in outcome, and another one did not account for any misclassification error. For models where misclassification error was not accounted for, an individual with any stool sample positive for a particular STH was considered as infection positive for that STH. For *S. japonicum*, epg per participant (intensity of infection) was obtained by averaging the epg of all stool samples collected from a participant, which is the most commonly used method for calculating overall epg per participant [1,39,40].

We assumed conditional independence between subsequent tests in our model, meaning in practice that when more than one sample was available from a subject, the test results are independent from each other, conditional on the person's true infection status. In other words, the probability of a positive (or negative) test depends only on the true status, and once this true status is known, does not depend on any test results from other days. This assumption seemed reasonable, and simplifies the statistical model compared to a model that might account for any between-day dependencies.

WinBUGS software (version 1.4.3, MRC Biostatistics Unit, Cambridge, UK) was used to implement the Gibbs sampler algorithm. Posterior medians of random samples derived from marginal posterior densities were used as point estimates, reported with 95% Bayesian credible intervals (BCI). The programs written in WinBUGS are available upon request to the authors.

Results

Of the 5624 individuals who agreed to participate in the study at baseline, 2276 (40.5%) constitute the group "at-risk". The "atrisk" group and those who were not treated with praziquantel or did not provided any stool sample during the follow-up ("not atrisk" group) are compared in Table 1. A higher proportion of people in the "at-risk" group had a positive schistosomiasis test at baseline (23.5%) as compared to those in the "not at-risk" group (10.5%). Because of this discrepancy, there were more rice farmers in the "at-risk" group than in the 'not at-risk' group (50.2% vs. 40.9%), since rice farming is associated with *S. japonicum* infection. Having been positive at baseline, however, did not have an impact on the probability of providing a stool sample at follow-up among those people who did receive treatment (75.9% vs. 76.1%).

Figure 1 displays the OR estimates for the exposure variable (STH infection) from models with and without correction for misclassification error. The OR estimates (95% BCI) for hookworm infection changed from 1.28 (0.93, 1.76) without any adjustment for misclassification error to 2.13 (1.16, 4.08) when both exposure (hookworm infection) and outcome (S. japonicum infection) were corrected for misclassification error. For A. lumbricoides and T. *trichiura*, the OR changed from 0.91 (0.66, 1.26) to 0.74 (0.43, 1.25) and 1.11 (0.80, 1.55) to 1.32 (0.80, 2.27), respectively. Correction for misclassification error in either exposure or outcome gave intermediate estimates. However, only adjusting for misclassification error in S. japonicum had a larger impact on the OR estimates and their 95% BCI than only adjusting for the misclassification error in the STH. In general, misclassification error-adjusted estimates were further away from the null value and had wider confidence intervals than non-adjusted estimates. In addition, the impact of adjusting for misclassification error on OR estimates and their 95% BCI was larger for hookworm which had the lowest sensitivity and specificity values.

Table 2 provides OR estimates for covariates from respective STH models, with and without adjustment for misclassification error. For all three STH models, misclassification error-unadjusted

Table 1. Characteristics of the individuals in the "at-risk" and "not at-risk" groups.

Characteristic	At-risk group, no. (%)	Not at-risk group, no. (%)		
N	2276 (40.5)	3348 (59.5)		
Age (years)				
<10	658 (28.9)	1193 (35.6)		
11–16	399 (17.5)	458 (13.7)		
17–40	618 (27.2)	1047 (31.3)		
>40	601 (26.4)	650 (19.4)		
Male	1274 (56.0)	1692 (50.5)		
Rice farming	1142 (50.2)	1368 (40.9)		
Positive schistosomiasis test at baseline	534 (23.5)	350 (10.5)		

Data collected on 5624 people living in 50 villages of Samar Province, the Philippines, 2003–2004. Individuals in the "at-risk" group were included in the analysis. Individuals in the "not at-risk" group provided stool sample at baseline but were not included in the analysis.

doi:10.1371/journal.pntd.0000995.t001

OR estimate for >40 year-old individuals (reference: ≤ 10 years) was approximately 1.5 times that found in the exposure and outcome misclassification-adjusted model. Also, for all three STH models, OR estimates for males (reference: females) from the misclassification error-adjusted model were considerably different from OR estimates found in the unadjusted model. In general, both exposure and outcome misclassification error-adjusted ORs, and only outcome-adjusted ORs were similar whereas misclassification error-unadjusted ORs and only exposure-adjusted ORs were similar. The estimated 95% BCI from models adjusting for misclassification error in the outcome variables, with or without adjustment for misclassification error of STH only did not impact the width of the 95% BCI of the ORs of other variables in the model.

Discussion

To our knowledge, this is the first longitudinal study to estimate the effect of STH infection on the 12-month risk of *S. japonicum* infection in a population where both of these infections are endemic. In addition, this study minimizes several potential biases by including adjustment for misclassification error in both dependent and independent variables, varying sensitivity and specificity of both tests depending on the numbers of samples available, accounting for clustering between individuals within villages, and taking care of other possible confounders. The adjusted model suggests that hookworm infection is associated with increased 12-month risk of *S. japonicum* infection following treatment with praziquantel. The two other STH studied did not have an important effect on the risk of infection with schistosomiasis.

Although our analysis included only about one third of the baseline participants from 50 villages, the longitudinal sample size was large enough for this analysis. When comparing individuals included in and excluded from the analysis, we found more rice farmers in the 'at-risk' group than in the 'not at-risk' group. This is because more males were treated than females (56.4% vs. 43.6%), and because more rice farmers were infected with *S. japonicum* at baseline. A larger proportion of individuals infected with *S. japonicum* at baseline received treatment [36]. However, this did not have an impact on the probability of providing a stool sample at follow-up among those people who did receive treatment. So, the use of the "at-risk" group of participants is unlikely to introduce selection bias and to affect the validity of our estimates.

Our results show that OR estimates for all three STHs are pulled away from the null value when the OR estimates are adjusted for misclassification error. This effect of non-differential misclassification has long been recognized, although this is not always the case when exposure and outcome variables are dependent, a discrete variable assumes more than two values, or there is misclassification error in the confounding variable [21,23,41,42].

The effect of misclassification on the OR estimates of the association between STH and the risk of *S. japonicum* infection differed for the three STHs under study. The magnitude of impact of misclassification error depends on the sensitivity, specificity, and true prevalence of the variable(s) of interest. The relative change in the OR estimates between the unadjusted model and the model adjusting for misclassification error of STH and *S. japonicum* was larger for hookworm than the other STHs. This is likely to be due to the considerably lower sensitivity (single stool sample) of the Kato-Katz for hookworm as compared to that for *A. humbricoides* and *T. trichiura* [43].

Two studies have reported estimates of cross-sectional association between hookworm infection and infection by another schistosome species (*S. mansoni*). Keiser and others (2002) reported



Figure 1. Odds ratio estimates for the exposure variable with and without correction for misclassification error. Exposure variable is respective soil-transmitted helminth infection; All odds ratio estimates are adjusted for age, sex, and occupation; BCI: Bayesian credible intervals; ME: misclassification error; ^a Correction of misclassification error in exposure (respective STH infection) and outcome (*S. japonicum* infection) assessment. doi:10.1371/journal.pntd.0000995.g001

an OR of 2.25 (95% CI: 1.31, 3.85) from their study conducted among 325 school children in Côte d'Ivoire [2]. Fleming and others (2006) reported an OR of 2.95 (95% CI: 2.19, 3.98) from a study conducted among 1332 individuals in Brazil [17]. Their results, which did not adjust for misclassification error, could be due to the cross-sectional nature of their study, which could increase the association between the prevalences of hookworm and schistomiasis. It is also possible that the association between hookworm and schistosomiasis is larger for S. mansoni than for S. japonicum or that the Kato-Katz performs better for the diagnosis of S. mansoni, thus reducing the effect of misclassifaction error. Moreover, temporality of the association could not be ascertained because of the cross-sectional design of these studies. Longitudinal design of our study allowed us to assess the impact of hookworm infection on the incidence of schistosomiasis japonica, after adequate adjustment for misclassification error. Even though the OR may overestimate somewhat the relative risk, these measures are likely to be reasonably close in our study since the risk of reinfection was in the order of 13%.

Important changes in OR estimates for other covariates were also observed. The OR estimates for covariates when only *S. japonicum* data (outcome) were adjusted for misclassification error were very close to the OR when both *S. japonicum* and STH data were adjusted. In contrast, the OR estimates for covariates when only STH data (exposure) were adjusted for misclassification error were very close to unadjusted OR estimates. This is because the strength of the association between the covariates and *S. japonicum* infection was considerably larger than the confounding effect of STH infections. Nevertheless, even correction for misclassification error in the outcome variable only was capable of changing estimate of effect of some of the covariates on the risk of *S. japonicum* infection. This has important implications for the assessment of the confounding effect of these variables and their association with the risk of *S. japonicum* infection. We also observed wider confidence intervals for all misclassification error-adjusted ORs. This results directly from incorporating uncertainty in estimating infection status [21,44].

The largest impact of misclassification error was observed for the association between hookworm and *S. japonicum*, which was negligible in the unadjusted model and important on the adjusted one. Several authors have provided numerical examples in their publications showing larger effects of joint misclassification of both exposure and outcome [22,41,45]. For *A. lumbricoides* and *T. trichiura*, OR point estimates indicate a negative and a positive relationship, respectively, but of a smaller magnitude.

The efficacy of praziguantel for the treatment of schistosomiasis has been reported to range between 71% and 99% in published literature [46,47,48]. However, more recent papers have reported an efficacy of praziquantel for the treatment of schistosomiasis around 96% [46,47]. The "at-risk" group size is likely to be affected by a lower efficacy as treatment with praziquantel does not completely cure everyone who has the infection. In our study, we assumed 100% efficacy of praziquantel for the treatment of schistosomiasis and decided not to adjust for a lower efficacy of praziquantal. This would have required yet another level of uncertainty for only a small proportion of the population (the efficacy is very high), and is unlikely to have changed our conclusions. Another limitation of this study is that our model assumes conditional independence of test results within each individual given the latent true infection status which is always uncertain. To assess conditional dependence we first have to build a more complex model assuming that there is at least some dependence. This allows examination of the size of the dependence parameter and whether or not its use is meaningful

Table 2. Odds ratio estimates for covariates from A. lumbricoides, Hookworm, and T. trichiura models.

	All ME correction ^a		Outcome ME correction		Exposure ME correction		No ME correction				
Covariates	OR	95% BCI	OR	95% BCI	OR	95% BCI	OR	95% BCI			
A. lumbricoides model											
Age	Reference: ≤10 yrs										
11–16 yrs	1.05	0.51, 2.11	1.05	0.50, 2.08	1.03	0.62, 1.67	1.04	0.63, 1.68			
17–40 yrs	0.49	0.22, 1.07	0.50	0.22, 1.04	0.60	0.34, 1.01	0.61	0.35, 1.03			
>40 yrs	0.21	0.09, 0.49	0.22	0.09, 0.49	0.36	0.20, 0.63	0.37	0.20, 0.65			
Sex	Reference	Reference: female									
Male	2.33	1.50, 3.75	2.33	1.48, 3.70	1.83	1.34, 2.50	1.84	1.35, 2.52			
Occupation	Reference	Reference: rice farming									
Non-rice farming	0.42	0.20, 0.86	0.42	0.20, 0.82	0.48	0.29, 0.78	0.49	0.30, 0.79			
Hookworm model											
Age	Reference	e: ≤10 yrs									
11–16 yrs	1.03	0.49, 2.07	1.07	0.52, 2.12	1.02	0.61, 1.66	1.03	0.62, 1.67			
17–40 yrs	0.44	0.19, 0.94	0.48	0.22, 1.00	0.56	0.32, 0.96	0.58	0.33, 1.01			
>40 yrs	0.20	0.08, 0.45	0.21	0.09, 0.47	0.34	0.18, 0.60	0.35	0.19, 0.63			
Sex	Reference	Reference: female									
Male	2.12	1.35, 3.34	2.15	1.38, 3.48	1.72	1.26, 2.37	1.75	1.28, 2.40			
Occupation	Reference: rice farming										
Non-rice farming	0.44	0.21, 0.87	0.45	0.22, 0.88	0.49	0.30, 0.79	0.50	0.30, 0.82			
T. trichiura model											
Age	Reference: ≤10 yrs										
11–16 yrs	1.09	0.52, 2.16	1.08	0.52, 2.14	1.03	0.62, 1.68	1.03	0.63, 1.68			
17–40 yrs	0.54	0.25, 1.15	0.54	0.24, 1.13	0.63	0.36, 1.07	0.62	0.36, 1.05			
>40 yrs	0.24	0.10, 0.53	0.24	0.10, 0.53	0.37	0.21, 0.66	0.37	0.21, 0.66			
Sex	Reference: female										
Male	2.30	1.48, 3.68	2.31	1.48, 3.66	1.82	1.34, 2.50	1.83	1.34, 2.50			
Occupation	Reference	Reference: rice farming									
Non-rice farming	0.43	0.21, 0.85	0.43	0.20, 0.84	0.49	0.30, 0.79	0.49	0.30, 0.79			

Odds ratio estimates were calculated using data collected on 2276 people living in 50 villages of Samar Province, the Philippines, 2003–2004; BCI: Bayesian credible intervals; ME: misclassification error; OR: odds ratio;

^aCorrection of misclassification error in exposure (respective STH infection) and outcome (*S. japonicum* infection) assessment.

doi:10.1371/journal.pntd.0000995.t002

[49]. Exploring such a complex model is beyond the scope of this paper. However, several authors have noted that overlooking conditional dependence does not substantially change parameter estimates [49–51]. Our results were adjusted for risk factors most often reported to be associated with schistosomiasis, and often shared with hookworm, such as age, gender, occupation, and the village where people live. Although some additional unmeasured confounding factors may explain the observed association, such factors would need to have a very strong relationship with both hookworm and schistosomiasis to modify our conclusion.

Our data suggest that hookworm infection is associated with increased 12-month cumulative incidence of *S. japonicum* infection. Such important associations might be missed in analyses which do not adjust for misclassification errors. Our findings have important implications for control of these infections in regions where these

References

 Ezeamama AE, Friedman JF, Olveda RM, Acosta LP, Kurtis JD, et al. (2005) Functional significance of low-intensity polyparasite helminth infections in anemia. J infect dis 192: 2160–2170. worms are co-endemic. Effective control of one helminth can lead to reduction in incidence of another and help to reduce the overall burden of helminthic infection in affected regions.

Supporting Information

Checklist S1 STROBE Checklist.

Found at: doi:10.1371/journal.pntd.0000995.s001 (0.09 MB DOC)

Author Contributions

Conceived and designed the experiments: MRT. Performed the experiments: HC STM EB RO. Analyzed the data: MRT HC IJ. Wrote the paper: MRT. Interpreted the data: MRT HC IJ. Contributed to all phases of the field study: STM. Supervised the acquisition of data: EB RO.

 Keiser J, N'Goran EK, Singer BH, Lengeler C, Tanner M, et al. (2002) Association between *Schistosoma mansoni* and hookworm infections among schoolchildren in Côte d'Ivoire. Acta Trop 84: 31–41.

- Ezeamama AE, McGarvey ST, Acosta LP, Zierler S, Manalo DL, et al. (2008) The synergistic effect of concomitant schistosomiasis, hookworm, and *Trichuris* infections on children's anemia burden. PLoS Negl Trop Dis 2: e245.
- 4. Hotez PJ, Bundy DA, Beegle K, Brooker S, Drake L, et al. (2006) Helminth infections: soil-transmitted helminth infections and schistosomiasis. In: Jamison DT, Breman JG, Measham AR, Alleyne G, Claeson M, et al. (2006) Disease control priorities in developing countries. 2nd ed. New York, NY: Oxford University Press. pp 467–482.
- Hotez PJ, Molyneux DH, Fenwick A, Ottesen E, Ehrlich Sachs S, et al. (2006) Incorporating a rapid-impact package for neglected tropical diseases with programs for HIV/AIDS, tuberculosis, and malaria. PLoS Med 3: e102.
- Belo S, Rompão H, Gonçalves L, Grácio MA (2005) Prevalence, behavioural and social factors associated with *Schistosoma intercalatum* and geohelminth infections in São Tomé and Principe. Parassitologia 47: 227–231.
- WHO (2002) Prevention and control of schistosomiasis and soil-transmitted helminthiasis: report of a WHO expert committee. WHealthO Tech Rep Ser 912: 1–57.
- Utzinger J, Müller I, Vounatsou P, Singer BH, N'Goran EK, et al. (2003) Random spatial distribution of *Schistosoma mansoni* and hookworm infections among school children within a single village. J Parasitol 89: 686–692.
- Maszle DR, Whitehead PG, Johnson RC, Spear RC (1998) Hydrological studies of schistosomiasis transport in Sichuan Province, China. Sci Total Environ 216: 193–203.
- Blas BL, Rosales MI, Lipayon IL, Yasuraoka K, Matsuda H, et al. (2004) The schistosomiasis problem in the Philippines: a review. Parasitol Int 53: 127–134.
- Matthys B, Tschannen AB, Tian-Bi NT, Comoé H, Diabaté S, et al. (2007) Risk factors for *Schistosoma mansoni* and hookworm in urban farming communities in western Côte d'Ivoire. Trop Med Int Health 12: 709–723.
- Fulford AJC, Webster M, Ouma JH, Kimani G, Dunne DW (1998) Puberty and age-related changes in susceptibility to schistosome infection. Parasitol Today 14: 23–26.
- Ross AGP, Li YS, Sleigh AS, Li Y, Williams GM, et al. (1997) Epidemiologic features of *Schistosoma japonicum* among fishermen and other occupational groups in the Dongting Lake region (Hunan Province) of China. Am J Trop Med Hyg 57: 302–308.
- Bethony J, Brooker S, Albonico M, Geiger SM, Loukas A, et al. (2006) Soiltransmitted helminth infections: ascariasis, trichuriasis, and hookworm. Lancet 367: 1521–1532.
- Cox FEG (2001) Concomitant infections, parasites and immune responses. Parasitology 122 Suppl: S23–38.
- Chamone M, Marques CA, Atuncar GS, Pereira AL, Pereira LH (1990) Are there interactions between schistosomes and intestinal nematodes? Trans R Soc Trop Med Hyg 84: 557–558.
- Fleming FM, Brooker S, Geiger SM, Caldas IR, Correa-Oliveira R, et al. (2006) Synergistic associations between hookworm and other helminth species in a rural community in Brazil. Trop Med Int Health 11: 56–64.
- de Cássia Ribeiro Silva R, Barreto ML, Assis AM, de Santana ML, Parraga IM, et al. (2007) The relative influence of polyparasitism, environment, and host factors on schistosome infection. Am J Trop Med Hyg 77: 672–675.
- Quinnell RJ, Pritchard DI, Raiko A, Brown AP, Shaw MA (2004) Immune responses in human necatoriasis: association between interleukin-5 responses and resistance to reinfection. J Infect Dis 190: 430–438.
- Olatunde BO, Onyemelukwe GC (1994) Immunosuppression in Nigerians with hookworm infection. Afr J Med Med Sci 23: 221–225.
- Kelsey JL, Whittemore AS, Evans AS, Thompson WD (1996) Methods in observational epidemiology. 2nd ed. New York, NY: Oxford University Press.
- Brenner H, Savitz DA, Gefeller O (1993) The effects of joint misclassification of exposure and disease on epidemiologic measures of association. J Clin Epidemiol 46: 1195–1202.
- Bashir SA, Duffy SW, Qizilbash N (1997) Repeat measurement of case-control data: corrections for measurement error in a study of ischaemic stroke and haemostatic factors. Int J Epidemiol 26: 64–70.
- Dendukuri N, Rahme E, Bélisle P, Joseph L (2004) Bayesian sample size determination for prevalence and diagnostic test studies in the absence of a gold standard test. Biometrics 60: 388–397.
- Joseph L, Gyorkos TW, Coupal L (1995) Bayesian estimation of disease prevalence and the parameters of diagnostic tests in the absence of a gold standard. Am J Epidemiol 141: 263–272.
- Yu JM, de Vlas SJ, Yuan HC, Gryseels B (1998) Variations in fecal Schistosoma japonicum egg counts. Am J Trop Med Hyg 59: 370–375.
- Hall A (1981) Quantitative variability of nematode egg counts in faeces: a study among rural Kenyans. Trans R Soc Trop Med Hyg 75: 682–687.
- Anderson RM, Schad GA (1985) Hookworm burdens and faecal egg counts: an analysis of the biological basis of variation. Trans R Soc Trop Med Hyg 79: 812–825.

- Misclassification and Epidemiological Associations
- Booth M, Vounatsou P, N'Goran EK, Tanner M, Utzinger J (2003) The influence of sampling effort and the performance of the Kato-Katz technique in diagnosing *Schistosoma mansoni* and hookworm co-infections in rural Côte d'Ivoire. Parasitology 127: 525–531.
- Utzinger J, N'Goran EK, Marti HP, Tanner M, Lengeler C (1999) Intestinal amoebiasis, giardiasis and geohelminthiases: their association with other intestinal parasites and reported intestinal symptoms. Trans R Soc Trop Med Hyg 93: 137–141.
- de Vlas SJ, Gryseels B (1992) Underestimation of *Schistosoma mansoni* prevalences. Parasitol Today 8: 274–277.
- Engels D, Sinzinkayo E, Gryseels B (1996) Day-to-day egg count fluctuation in Schistosoma mansoni infection and its operational implications. Am J Trop Med Hyg 54: 319–324.
- McGarvey ST, Carabin H, Balolong E, Jr., Bélisle P, Fernandez T, et al. (2006) Cross-sectional associations between intensity of animal and human infection with *Schistosoma japonicum* in Western Samar province, Philippines. Bull World Health Organ 84: 446–452.
- Olds GR, King C, Hewlett J, Olveda R, Wu G, et al. (1999) Double-blind placebo-controlled study of concurrent administration of albendazole and praziquantel in schoolchildren with schistosomiasis and geohelminths. J Infect Dis 179: 996–1003.
- Peters PA, El AM, Warren KS, Mahmoud AAF (1980) Quick Kato smear for field quantification of *Schistosoma mansoni* eggs. Am J Trop Med Hyg 29: 217–219.
- Tallo VL, Carabin H, Alday PP, Balolong E, Jr., Olveda RM, et al. (2008) Is mass treatment the appropriate schistosomiasis elimination strategy? Bull World Health Organ 86: 765–771.
- 37. Carabin H, Balolong E, Joseph L, McGarvey ST, Johansen MV, et al. (2005) Estimating sensitivity and specificity of a faecal examination method for *Schistosoma japonicum* infection in cats, dogs, water buffaloes, pigs, and rats in Western Samar and Sorsogon Provinces, The Philippines. Int J Parasitol 35: 1517–1524.
- Tarafder MR, Balolong E, Jr., Carabin H, Bélisle P, Tallo V, et al. (2006) A cross-sectional study of the prevalence of intensity of infection with *Schistosoma japonicum* in 50 irrigated and rain-fed villages in Samar Province, the Philippines. BMC Public Health 6: 61.
- Knopp S, Mgeni AF, Khamis IS, Steinmann P, Stothard JR, et al. (2008) Diagnosis of soil-transmitted helminths in the era of preventive chemotherapy: effect of multiple stool sampling and use of different diagnostic techniques. PLoS Negl Trop Dis 2: e331.
- Zhou YB, Yang MX, Wang QZ, Zhao GM, Wei JG, et al. (2007) Field comparison of immunodiagnostic and parasitological techniques for the detection of schistosomiasis japonica in the People's Republic of China. Am J Trop Med Hyg 76: 1138–1143.
- Kleinbaum DG, Kupper LL, Morgenstern H (1982) Epidemiologic research: priciples and quantitative methods. 1st ed. New York, NY: John Wiley and Sons.
- Marshall JR, Hastrup JL (1996) Mismeasurement and the resonance of strong confounders: uncorrelated errors. Am J Epidemiol 143: 1069–1078.
- 43. Tarafder MR, Carabin H, Joseph L, Balolong E, Jr., Olveda R, et al. (2010) Estimating the sensitivity and specificity of Kato-Katz stool examination technique for detection of hookworms, *Ascaris lumbricoides* and *Trichuris trichiura* infections in humans in the absence of a 'gold standard'. Int J Parasitol 40: 399–404.
- Höfler M (2005) The effect of misclassification on the estimation of association: a review. Int J Methods Psychiatr Res 14: 92–101.
- Vogel C, Brenner H, Pfahlberg A, Gefeller O (2005) The effects of joint misclassification of exposure and disease on the attributable risk. Stat Med 24: 1881–1896.
- Grover JK, Vats V, Uppal G, Yadav S (2001) Anthelmintics: a review. Trop Gastroenterol 22: 180–189.
- 47. Hou XY, McManus DP, Gray DJ, Balen J, Lou XS, et al. (2008) A randomized, double-blind, placebo-controlled trial of safety and efficacy of combined praziquantel and artemether treatment for acute schistosomiasis japonica in China. Bull World Health Organ 86: 788–795.
- King CH, Mahmoud AAF (1989) Drugs five years later: praziquantel. Ann Intern Med 110: 290–296.
- Dendukuri N, Joseph L (2001) Bayesian approaches to modeling the conditional dependence between multiple diagnostic tests. Biometrics 57: 158–167.
- Black MA, Craig BA (2002) Estimating disease prevalence in the absence of a gold standard. Stat Med 21: 2653–2669.
- Gustafson P (2005) On model expansion, model contraction, identifiability and prior information: two illustrative scenarios involving mismeasured variables. Stat Sci 20: 111–140.